## UC San Diego UC San Diego Previously Published Works

### Title

Geohealth Policy Benefits Are Mediated by Interacting Natural, Engineered, and Social Processes.

Permalink https://escholarship.org/uc/item/0rq7x0qq

**Journal** GeoHealth, 7(9)

**Authors** Calder, Ryan Schartup, Amina

Publication Date 2023-09-01

## DOI

10.1029/2023GH000858

Peer reviewed

## GeoHealth

#### COMMENTARY

10.1029/2023GH000858

#### **Key Points:**

- Policy advice in geohealth must acknowledge health and non-health tradeoffs
- Health is governed by complex social as well as natural/engineered systems
- Funding agencies can improve the quality of policy advice from geohealth by allowing or promoting a health nexus in funded projects

Correspondence to:

R. S. D. Calder, rsdc@vt.edu

#### Citation:

Calder, R. S. D., & Schartup, A. T. (2023). Geohealth policy benefits are mediated by interacting natural, engineered, and social processes. *GeoHealth*, 7, e2023GH000858. https:// doi.org/10.1029/2023GH000858

Received 15 MAY 2023 Accepted 17 AUG 2023

#### **Author Contributions:**

Conceptualization: Ryan S. D. Calder, Amina T. Schartup Formal analysis: Ryan S. D. Calder, Amina T. Schartup Investigation: Ryan S. D. Calder, Amina T. Schartup Methodology: Ryan S. D. Calder, Amina T. Schartup Visualization: Ryan S. D. Calder, Amina T. Schartup Writing – original draft: Ryan S. D. Calder, Amina T. Schartup Writing – review & editing: Ryan S. D. Calder, Amina T. Schartup

© 2023 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# and social mechanisms, and health and non-health tradeoffs, is needed to achieve policy benefits rather than "policy impact." Dedicated funding mechanisms would improve the quality and availability of these analyses.

Oceanography, La Jolla, CA, USA

**Plain Language Summary** The Earth sciences can produce information of direct relevance to human health, for example, on the cycling of contaminants. Researchers provide policy advice in published articles with the goal of reducing health impacts. Many proposed policies, such as food consumption advisories, introduce complex tradeoffs that are rarely directly analyzed. Others, such as investments in pollution control technologies, have opportunity costs. This article encourages researchers to interrogate possible unintended consequences or tradeoffs. This is likely to involve social as well as natural or engineered processes. This article identifies possible barriers to effectively implementing this thinking, notably, funding mechanisms that exclude study of health endpoints.

**Geohealth Policy Benefits Are Mediated by Interacting** 

<sup>1</sup>Department of Population Health Sciences, Virginia Tech, Blacksburg, VA, USA, <sup>2</sup>Global Change Center, Virginia

Tech, Blacksburg, VA, USA, <sup>3</sup>Faculty of Health Sciences, Virginia Tech, Roanoke, VA, USA, <sup>4</sup>Department of Civil and

**Abstract** Interest in health implications of Earth science research has significantly increased. Articles frequently dispense policy advice, for example, to reduce human contaminant exposures. Recommendations such as fish consumption advisories rarely reflect causal reasoning around tradeoffs or anticipate how scientific

information will be received and processed by the media or vulnerable communities. Health is the product of

interacting social and physical processes, yet predictable responses are often overlooked. Analysis of physical

Environmental Engineering, Duke University, Durham, NC, USA, 5Geosciences Research Division, Scripps Institution of

Natural, Engineered, and Social Processes

Ryan S. D. Calder<sup>1,2,3,4</sup> (D) and Amina T. Schartup<sup>5</sup> (D)

#### 1. Introduction

Recent years have seen great increase in interest in the health implications of Earth science research among geoscientists, policymakers, and others. Of all geochemistry-geophysics papers indexed by Web of Science since 1987 mentioning "health" in the abstract, roughly half have been published since 2015. This interest has been reflected in the proliferation of publications (e.g., GeoHealth by the American Geophysical Union) and professional societies (e.g., Geology for Global Development) devoted to mobilizing Earth science expertise to improve human health and wellbeing. This research has the potential to be highly convergent (NRC, 2014), linking physical models developed at large spatial and temporal scales (e.g., multi-decadal global climate models) to individual- or community-scale socio-environmental models for human exposures and health outcomes.

Geohealth research products most often characterize a natural process (e.g., environmental contaminant cycling) and make inferences about risks to human health by drawing on the health sciences literature. Policy recommendations tend to represent humans as receptors whose overall health will be improved or impaired by simple policy or behavioral levers. For example, the geohealth literature is replete with calls for fish consumption advisories, emissions controls, capital investments, and other interventions intended to achieve a human health benefit. Implicit in these recommendations is a single causal chain between natural system, human exposures, and health outcomes. Whether or not proposed interventions would minimize human health impacts, cost-effectiveness, feasibility, or potential tradeoffs are rarely directly interrogated in articles providing policy advice.

Advice on increasing the "policy impact" of environmental research has traditionally focused on study design and publication practices (Bilotta et al., 2015) and has even encouraged authors to recast policy debates around existing research agendas (Rose et al., 2020). This is problematic because it reinforces monodisciplinary models of health and the environment and encourages decision-making based on the persuasive power of individual scientists or research groups. In reality, net health impacts of policy interventions are determined by interacting natural, engineered, and social processes, which are currently poorly represented in geohealth research. Scientific

#### AGGU ADVANCING EARTH AND SPACE SCIENCES







**Figure 1.** Earth science investigations frequently conclude with policy advice that anticipates a straightforward impact on human health (black arrows). We call for analysis that considers the diverse mechanisms by which advice may pose health tradeoffs instead (white arrows). The conventional scientific workflow (top left) can also be enhanced by greater knowledge co-production and other strategies discussed in Section 4.

research thus has greater policy *benefit* when it provides a structured analysis of tradeoffs actually confronted by policymakers, which frequently cross traditional disciplinary lines, and when it seeks to "inform, not persuade" (Blastland et al., 2020).

In Figure 1, we illustrate how policy advice may emerge as a byproduct of the traditional scientific workflow in the Earth sciences. In the United States, policy insights from geoscience work frequently emerge as co-benefits of traditional scientific activities, for example, as a "broader impact" in the terminology of the National Science Foundation. Findings from a well-defined geoscience hypothesis are often interpreted in the context of risks to human health, and corresponding advice is issued to reduce those risks. Ultimately, we call for geohealth research that identifies potential policy interventions carefully and with consideration of the range of likely mechanisms that could pose, at a minimum, health *tradeoffs*, if not tradeoffs across health and non-health outcomes. A full policy analysis is understandably outside the scope of most Earth science research. However, a lack of causal reasoning around proposed policy interventions may lead to adverse consequences for communities who may otherwise benefit most from scientific research.

There are many examples of adverse outcomes from geohealth interventions linked to inadequate consideration of the interacting social, natural, or engineered processes in which they are embedded. These include rebound effects, where efficiency improvements designed to arrest depletion of environmental resources instead accelerate their depletion by lowering costs (Berbel et al., 2014; Wheeler et al., 2020); the adverse impacts of green infrastructure on ground-level air quality via perturbations to local atmospheric chemistry and physics (Wang, 2021; J. Zhang et al., 2019; Y. Zhang et al., 2020); changes to drinking water sources made to avoid pathogen exposures or reduce costs increasing exposures to other contaminants (Escamilla et al., 2011; Masten et al., 2016); and fish consumption advisories that reduce contaminant exposures but also nutrient intakes, and which often affect dietary patterns of individuals other than those targeted (Calder et al., 2019; Shimshack et al., 2007; Stone & Hope, 2010).

We use the example of fish and seafood consumption advisories as a case study to demonstrate how seemingly simple policy advice is complicated by tradeoffs across life science disciplines and uncertainties across natural and social systems. We continue with a generalizable discussion of how geohealth research is complicated by patterns of information diffusion, including the role of the media. We close with a set of recommendations to scientists and policymakers.

#### 2. Case Study: Fish and Seafood Consumption Advisories

Fish and seafood bioaccumulate chemicals such as polychlorinated biphenyls and methylmercury (MeHg), and consumption advisories are a commonly advocated and deployed tool to control human exposures (Fair et al., 2018; Ghoshdastidar et al., 2018; Imm et al., 2005; Li et al., 2022; Ni et al., 2021; Sundseth et al., 2015).

Yet fish and seafood are also low-fat, high-protein foods rich in micronutrients, and there is a rich evidence base for the cardiovascular and neurodevelopmental health benefits of fish and seafood consumption (Ruxton, 2011). In practice, consumption advisories are designed to contain human exposures to contaminants within the range of reference doses (levels judged unlikely to result in excess risk) or, for carcinogenic/genotoxic contaminants, a dose associated with an acceptable excess lifetime cancer risk (often 1 in 100,000) (U.S. EPA, 2000).

Interventions to lower contaminant exposures to a risk-based target are not necessarily health-protective if there is a countervailing impact on nutrient intake, particularly if baseline contaminant exposures are near or below reference doses (Hellberg et al., 2012; Stone & Hope, 2010). For example, MeHg and  $\omega$ -3 long-chain polyunsaturated fatty acids ( $\omega$ -3 PUFAs) from fish and seafood exert opposite effects on common cellular mechanisms that influence cardiovascular and neurodevelopmental risks such that most fish are likely to present net benefits over wide ranges of intakes (Ginsberg & Toal, 2009; U.S. FDA, 2014). However, risk-benefit models vary widely in assumptions over plateaus in benefits, dose-response shapes for both MeHg and  $\omega$ -3 PUFAs, the range of endpoints considered, approaches to weight the importance of each endpoint, and assumptions about counterfactual dietary patterns (Calder et al., 2019; Groth, 2017; Rice et al., 2010; Stern & Korn, 2011). Thus, even idealized models of likely health impacts of changes in fish consumption necessarily embed subjective professional judgments of the strength of diverse lines of epidemiological evidence and the relative importance of different risks and benefits.

These uncertainties are compounded by unpredictable responses to fish consumption advisories. Targeted individuals may be unaware of advisories (Anderson et al., 2004) or ignore them (Burger, 2000). Conversely, advisories can also lead to reduced consumption of species not targeted by an advisory and/or among individuals not within the scope of the advisory (Shimshack et al., 2007). Among individuals who reduce consumption of fish and seafood, net health benefits are extremely sensitive to what foods are substituted in diets; even modest increases in red meat consumption are likely to have significant net negative effects on cancer and cardiovascular risks, and there are substantial uncertainties associated with patterns of diet switching (Calder et al., 2019; Cohen et al., 2005). There is a literature on improving the understanding of consumption advisories (Burger, 2000; Tan et al., 2011), but ultimately there is no predictive model for the social responses to these interventions.

Overall, calls for fish and seafood consumption advisories, like other geohealth interventions, are only justified following a comparison of risks of action versus inaction. This calculation is often made successfully in cases of urgent hazards with well-understood solutions (e.g., changing water sources, repairing treatment infrastructure, or distributing bottled water following an outbreak of waterborne disease). Conversely, smaller risks are often understood using inferential techniques such as low-dose extrapolation, introducing substantial uncertainties and subjective elements that complicate a comparison with the risks of intervention. Many geohealth interventions anticipate a response from individuals, but these are unpredictable and interact with uncertainties in natural or engineered systems in subtle ways as the above case study demonstrates.

#### 3. Geohealth Is Complicated by Patterns of Information Diffusion

Geohealth interventions are complicated by information diffusion patterns beyond the control of scientists, notably, among the media, individuals within affected communities, and groups with an interest in scientific findings. In most sciences, individual researchers have great latitude to decide how much to engage with the media, the public, or policymakers as a function of their own perceptions, incentives, priorities, and aptitudes (Besley & Nisbet, 2013; Corley et al., 2011). Conversely, geohealth research often addresses questions of high public salience and may spawn media coverage and sustained discussion in affected communities; in these cases, researchers may have a responsibility to engage with media to ensure coverage and perceptions are consistent with actual scientific findings.

Research with a health nexus receives vastly more media attention than other areas of scientific inquiry, for example, >90% of U.S. mass media articles in 1990 and 2001 (Suleski & Ibaraki, 2009). Traditional and social media coverage drives awareness of environmental risks in settings ranging from Indigenous communities in Canada (Furgal et al., 2005; Wheatley & Paradis, 1996) to the general populations of diverse countries where studies have been conducted, including the U.S., Italy, and China (Maran & Begotti, 2021; Ragain, 2009; Tilt & Qing, 2010). Scientists have little control over media representation or interpretation of published findings, which can lead to unexpected complications: media coverage of health research is prone to sensationalism, can overstate confidence

in, and generalizability of, findings, and tends to focus on adversarial dynamics of scientific debates rather than critically evaluating the strength of evidence on either side (Chang, 2015; Nabi & Prestin, 2016; Shuchman & Wilkes, 1997). Implications for individual decision-making (e.g., whether to eat less fish) presented by research articles may thus be magnified or distorted by media coverage regardless of the intent of the authors. Authors may minimize the potential for misrepresentation of research findings by actively engaging with media and addressing and confronting likely misunderstandings directly with reporters.

The communities most vulnerable to environmental hazards tend to be the ones most vulnerable to incomplete, misleading, or alarmist information. For example, Indigenous communities in the Arctic face high contaminant exposures due to patterns of global contaminant cycling and deposition and diets rich in predatory fish and mammals (Bard, 1999; Van Oostdam et al., 1999). Communicating scientific information on environmental risks to Indigenous communities is however complicated by language barriers, differences in conception of risk, and the potential for distortion of scientific information across large social networks (Arquette et al., 2002; Furgal et al., 2005; Mercer et al., 2010). Individuals from different backgrounds may have attachments to foods, traditional medicines, and other cultural products that represent genuine hazards (e.g., heavy metals in traditional cosmetics) but that also fulfill culturally specific psychosocial roles (Ernst, 2002; King & Furgal, 2014; U.S. EPA, 2023). Potential health gains from interventions in culturally specific practices should be weighed against impacts of disruption, ideally, in close collaboration with representatives from the affected communities (IARPC, 2018).

Finally, geohealth findings and policy advice may be used in unanticipated ways by special interest groups. For example, our own work characterizing health impacts of new hydroelectric reservoirs in Canada has been cited by U.S. environmental groups opposed to expanding cross-border electrical interties (Birchard, 2017; Calder et al., 2016; Riverkeeper Inc., 2019). Because such interties usually draw from existing rather than new generation, different environmental risks and benefits are relevant to the policy debate around intertie expansion. We thus did not anticipate that our research would be used to lobby against deeper integration of the electrical grid, which is widely considered to be a key pathway to decarbonization (Dimanchev et al., 2021). We have since countered these arguments through media outreach and an expanded research agenda that characterizes social costs and benefits of hydropower exports in the broader context of the renewable energy transition (Calder, 2019a, 2019b; Calder et al., 2020b, 2022; Matson, 2022; Willson, 2023).

#### 4. Strategies to Increase the Policy Benefits of Geohealth Research

The convergent nature of geohealth research creates unique opportunities to use scientific analysis to support decision-making around complex policy challenges. However, as described above, policy recommendations that do not adequately reflect the interactions of social, natural, and engineered systems, or that do not appropriately reflect uncertainties, may be counterproductive.

Knowledge co-production has emerged as a set of practices to build communities affected by environmental (and other) risks into the research process, from hypothesis generation to articulation of scientific findings, and, if justified by the scope of the analysis, policy advice (Armitage et al., 2011; Woodall et al., 2021). Early and active involvement of members of the public, community activists, policy experts, and/or other stakeholders from a variety of backgrounds increases the likelihood that research will reflect relevant tradeoffs. Funding the participation of community members and organizations to the extent possible is crucial given their time and resource constraints. This is especially the case for organizations serving historically marginalized or disadvantaged communities, which may face the greatest constraints while engaged with the most vulnerable groups.

There is a growing literature providing guidance on community-engaged science, with models varying according to timeline, whether a project is community-led or researcher-led, and whether the primary outcome of interest is generalizable knowledge or site-specific decision support or interventions (David-Chavez & Gavin, 2018; Hayhow et al., 2021). This adds to emerging guidance from the U.S. Government, for example, the Interagency Arctic Research Policy Committee Principles for Conducting Research in the Arctic (IARP, 2018). These strategies are essential for creating informative and policy-relevant science and also for using science to meaningfully advance environmental justice (Hoffman-Hall et al., 2022).

Graphical modeling tools, known variously as "results chains," "logic models," "Bayesian networks," and other terms, have emerged as a set of methods to structure relationships among interventions and responses among

social and physical systems (Calder et al., 2020a; Tallis et al., 2019). Graphical models can be used to identify potential synergistic or antagonistic processes and areas of uncertainty in order to inform new research questions or identify the possibility for unintended consequences. Development of these models also encourages interdisciplinary collaboration and community engagement to ensure that all relevant pathways and concepts have been incorporated. Policy advice based on a detailed graphical model is likely to be more nuanced and comprehensive than one that emerges as a byproduct of a relatively narrow geoscience investigation.

Graphical models have found some use in international development and other social sciences, in environmental health, and in the rapidly growing field of planetary health, but are still rare in the Earth sciences. Interest in these methods may rise after COVID-19 demonstrated the need to balance spread of infectious disease with countervailing economic, psychological, and other impacts of concern (Alexander & Shareck, 2021; Calder et al., 2021; Diallo et al., 2023). Overall, there is a growing literature from which geoscientists can draw to better align scientific research with policy impact to the benefit of communities facing environmental risks and policymakers facing poorly elucidated tradeoffs.

Barriers to effective practice of convergent geohealth remain. Perhaps most significantly, geoscience and health research funding is directed overwhelmingly at basic science, while policy transferability and community engagement are often considered a "broader impact," at least in the United States (Figure 1) (Seitter et al., 2022). A recent analysis by the American Meteorological Society concluded that research advancing best practices in U.S. geohealth is likely to fall outside the scope of the primary federal funders of both geoscience (National Science Foundation) and health (National Institutes of Health) (Seitter et al., 2022). Likewise, major funders of interdisciplinary research often specifically exclude research with a health nexus (Alfred P. Sloan Foundation, 2022). Funding opportunities dedicated to convergent geohealth research such as NASA's Health and Air Quality program are rare (NASA, 2021). We note positive developments, for example, a recent NSF call for applications to supplemental funding for graduate students whose work links geoscience to human health (NSF, 2023). Still, despite significant emerging interest in geohealth, it is often pursued, by necessity, on an ad-hoc basis as allowed by available time and money. The lack of institutional structure around geohealth research is almost certainly impairing development of consensus around research priorities and the dissemination of best practices.

#### **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### **Data Availability Statement**

Data were not used, nor created for this research.

#### References

Alexander, S. A., & Shareck, M. (2021). Widening the gap? Unintended consequences of health promotion measures for young people during COVID-19 lockdown. *Health Promotion International*, *36*(6), 1783–1794. https://doi.org/10.1093/heapro/daab015

- Alfred P. Sloan Foundation. (2022). Call for letters of inquiry: Collaborative research projects on energy system electrification.
- Anderson, H. A., Hanrahan, L. P., Smith, A., Draheim, L., Kanarek, M., & Olsen, J. (2004). The role of sport-fish consumption advisories in mercury risk communication: A 1998-1999 12-state survey of women age 18-45. *Environmental Research*, 95(3), 315–324. https://doi. org/10.1016/j.envres.2004.01.004
- Armitage, D., Berkes, F., Dale, A., Kocho-Schellenberg, E., & Patton, E. (2011). Co-management and the co-production of knowledge: Learning to adapt in Canada's Arctic. *Global Environmental Change*, 21(3), 995–1004. https://doi.org/10.1016/j.gloenvcha.2011.04.006
- Arquette, M., Cole, M., Cook, K., LaFrance, B., Peters, M., Ransom, J., et al. (2002). Holistic risk-based environmental decision making: A native perspective. *Environmental Health Perspectives*, 110(Suppl 2), 259–264. https://doi.org/10.1289/ehp.02110s2259
- Bard, S. M. (1999). Global transport of anthropogenic contaminants and the consequences for the Arctic marine ecosystem. *Marine Pollution Bulletin*, 38(5), 356–379. https://doi.org/10.1016/s0025-326x(99)00041-7

Berbel, J., Gutiérrez-Martín, C., Rodríguez-Díaz, J. A., Camacho, E., & Montesinos, P. (2014). Literature review on rebound effect of water saving measures and analysis of a Spanish case study. Water Resources Management, 29(3), 663–678. https://doi.org/10.1007/s11269-014-0839-0

- Besley, J. C., & Nisbet, M. (2013). How scientists view the public, the media and the political process. *Public Understanding of Science*, 22(6), 644–659. https://doi.org/10.1177/0963662511418743
- Bilotta, G. S., Milner, A. M., & Boyd, I. L. (2015). How to increase the potential policy impact of environmental science research. *Environmental Sciences Europe*, 27(1), 9. https://doi.org/10.1186/s12302-015-0041-x
- Birchard, M. (2017). Re: Supplemental comments of CLF on DEIS and SDEIS, northern pass transmission LLC, presidential permit application, OE Docket No. PP-371. Conservation Law Foundation.
- Blastland, M., Freeman, A. L. J., van der Linden, S., Marteau, T. M., & Spiegelhalter, D. (2020). Five rules for evidence communication. *Nature*, 587(7834), 362–364. https://doi.org/10.1038/d41586-020-03189-1

#### Acknowledgments

Dr. Ryan Calder gratefully acknowledges support from NASA (Grant 80NSSC22K1048), and Dr. Amina Schartup gratefully acknowledges support from the National Science Foundation (Grant OCE 2023046).

- Burger, J. (2000). Consumption advisories and compliance: The fishing public and the deamplification of risk. Journal of Environmental Planning and Management, 43(4), 471–488. https://doi.org/10.1080/713676577
- Calder, R. S. D. (2019a). Canada ignores Muskrat Falls at its own peril. The Telegram.
- Calder, R. S. D. (2019b). Une mauvaise publicité pour l'hydroélectricité québécoise. La Presse.
- Calder, R. S. D., Alatorre, A., Marx, R. S., Mallampalli, V., Mason, S. A., Olander, L. P., et al. (2020a). Graphical models and the challenge of evidence-based practice in development and sustainability. *Environmental Modelling & Software*, 130, 104734. https://doi.org/10.1016/j. envsoft.2020.104734
- Calder, R. S. D., Bromage, S., & Sunderland, E. M. (2019). Risk tradeoffs associated with traditional food advisories for Labrador Inuit. Environmental Research, 168, 496–506. https://doi.org/10.1016/j.envres.2018.09.005
- Calder, R. S. D., Grady, C., Jeuland, M., Kirchhoff, C. J., Hale, R. L., & Muenich, R. L. (2021). COVID-19 reveals vulnerabilities of the food-energy-water nexus to viral pandemics. *Environmental Science and Technology Letters*, 8(8), 606–615. https://doi.org/10.1021/acs. estlett.1c00291
- Calder, R. S. D., Robinson, C. S., & Borsuk, M. E. (2020b). Analysis of environmental and economic impacts of hydropower imports for New York City through 2050. Duke University.
- Calder, R. S. D., Robinson, C. S., & Borsuk, M. E. (2022). Total social costs and benefits of long-distance hydropower transmission. *Environmen*tal Science & Technology, 56(24), 17510–17522. https://doi.org/10.1021/acs.est.2c06221
- Calder, R. S. D., Schartup, A. T., Li, M., Valberg, A. P., Balcom, P. H., & Sunderland, E. M. (2016). Future impacts of hydroelectric power development on methylmercury exposures of Canadian indigenous communities. *Environmental Science & Technology*, 50(23), 13115–13122. https://doi.org/10.1021/acs.est.6b04447
- Chang, C. (2015). Inaccuracy in health research news: A typology and predictions of scientists' perceptions of the accuracy of research news. Journal of Health Communication, 20(2), 177–186. https://doi.org/10.1080/10810730.2014.917746
- Cohen, J. T., Bellinger, D. C., Connor, W. E., Kris-Etherton, P. M., Lawrence, R. S., Savitz, D. A., et al. (2005). A quantitative risk-benefit analysis of changes in population fish consumption. *American Journal of Preventive Medicine*, 29(4), 325–334. https://doi.org/10.1016/j. amepre.2005.07.003
- Corley, E. A., Kim, Y., & Scheufele, D. A. (2011). Leading US nano-scientists' perceptions about media coverage and the public communication of scientific research findings. *Journal of Nanoparticle Research*, 13(12), 7041–7055. https://doi.org/10.1007/s11051-011-0617-3
- David-Chavez, D. M., & Gavin, M. C. (2018). A global assessment of Indigenous community engagement in climate research. Environmental Research Letters, 13(12), 123005. https://doi.org/10.1088/1748-9326/aaf300
- Diallo, I., Ndejjo, R., Leye, M. M. M., Egbende, L., Tusubira, A., Bamgboye, E. A., et al. (2023). Unintended consequences of implementing non-pharmaceutical interventions for the COVID-19 response in Africa: Experiences from DRC, Nigeria, Senegal, and Uganda. *Globalization* and Health, 19(1), 36. https://doi.org/10.1186/s12992-023-00937-6
- Dimanchev, E. G., Hodge, J. L., & Parsons, J. E. (2021). The role of hydropower reservoirs in deep decarbonization policy. *Energy Policy*, 155, 112369. https://doi.org/10.1016/j.enpol.2021.112369
- Ernst, E. (2002). Heavy metals in traditional Indian remedies. European Journal of Clinical Pharmacology, 57(12), 891–896. https://doi. org/10.1007/s00228-001-0400-y
- Escamilla, V., Wagner, B., Yunus, M., Streatfield, P. K., van Geen, A., & Emch, M. (2011). Effect of deep tube well use on childhood diarrhoea in Bangladesh. Bulletin of the World Health Organization, 89(7), 521–527. https://doi.org/10.2471/BLT.10.085530
- Fair, P. A., White, N. D., Wolf, B., Arnott, S. A., Kannan, K., Karthikraj, R., & Vena, J. E. (2018). Persistent organic pollutants in fish from Charleston Harbor and tributaries, South Carolina, United States: A risk assessment. *Environmental Research*, 167, 598–613. https://doi. org/10.1016/j.envres.2018.08.001
- Furgal, C. M., Powell, S., & Myers, H. (2005). Digesting the message about contaminants and country foods in the Canadian North: A review and recommendations for future research and action. Arctic, 58(2), 103–114. https://doi.org/10.14430/arctic404
- Ghoshdastidar, A. J., Hu, Z., Nazarenko, Y., & Ariya, P. A. (2018). Exposure to nanoscale and microscale particulate air pollution prior to mining development near a northern indigenous community in Quebec, Canada. *Environmental Science and Pollution Research*, 25(9), 8976–8988. https://doi.org/10.1007/s11356-018-1201-5
- Ginsberg, G. L., & Toal, B. F. (2009). Quantitative approach for incorporating methylmercury risks and omega-3 fatty acid benefits in developing species-specific fish consumption advice. *Environmental Health Perspectives*, 117(2), 267–275. https://doi.org/10.1289/ehp.11368
- Groth, E. (2017). Scientific foundations of fish-consumption advice for pregnant women: Epidemiological evidence, benefit-risk modeling, and an integrated approach. *Environmental Research*, 152, 386–406. https://doi.org/10.1016/j.envres.2016.07.022
- Hayhow, C. M., Brabander, D. J., Jim, R., Lively, M., & Filippelli, G. M. (2021). Addressing the need for just GeoHealth engagement: Evolving models for actionable research that transform communities, *GeoHealth*, 5(12), e2021GH000496, https://doi.org/10.1029/2021GH000496
- Hellberg, R. S., DeWitt, C. A. M., & Morrissey, M. T. (2012). Risk-benefit analysis of seafood consumption: A review. Comprehensive Reviews in Food Science and Food Safety, 11(5), 490–517. https://doi.org/10.1111/j.1541-4337.2012.00200.x
- Hoffman-Hall, A., Gorris, M. E., Anenberg, S., Bredder, A. E., Dhaliwal, J. K., Diaz, M. A., et al. (2022). A GeoHealth call to action: Moving beyond identifying environmental injustices to co-creating solutions. *GeoHealth*, 6(11), e2022GH000706. https://doi.org/10.1029/2022GH000706
- Imm, P., Knobeloch, L., Anderson, H. A., & Consortium, t. G. L. S. F. (2005). Fish consumption and advisory awareness in the Great Lakes Basin. *Environmental Health Perspectives*, 113(10), 1325–1329. https://doi.org/10.1289/ehp.7980
- Interagency Arctic Research Policy Committee (IARPC). (2018). Principles for conducting research in the Arctic. National Science Foundation. King, U., & Furgal, C. (2014). Is hunting still healthy? Understanding the interrelationships between indigenous participation in land-based
- practices and human-environmental health. International Journal of Environmental Research and Public Health, 11(6), 5751–5782. https:// doi.org/10.3390/ijerph110605751
- Li, A., Tang, Q., Kearney, K. E., Nagy, K. L., Zhang, J., Buchanan, S., & Turyk, M. E. (2022). Persistent and toxic chemical pollutants in fish consumed by Asians in Chicago, United States. *Science of the Total Environment*, 811, 152214. https://doi.org/10.1016/j.scitotenv.2021.152214 Maran, D. A., & Begotti, T. (2021). Media exposure to climate change, anxiety, and efficacy beliefs in a sample of Italian university students.
- International Journal of Environmental Research and Public Health, 18(17), 9358. https://doi.org/10.3390/ijerph18179358
- Masten, S. J., Davies, S. H., & McElmurry, S. P. (2016). Flint water crisis: What happened and why? Journal of the American Water Works Association, 108(12), 22–34. https://doi.org/10.5942/jawwa.2016.108.0195
- Matson, Z. (2022). Power through the park. Adirondack Explorer.
- Mercer, J., Kelman, I., Taranis, L., & Suchet-Pearson, S. (2010). Framework for integrating indigenous and scientific knowledge for disaster risk reduction. *Disasters*, 34(1), 214–239. https://doi.org/10.1111/j.1467-7717.2009.01126.x
- Nabi, R. L., & Prestin, A. (2016). Unrealistic hope and unnecessary fear: Exploring how sensationalistic news stories influence health behavior motivation. *Health Communication*, 31(9), 1115–1126. https://doi.org/10.1080/10410236.2015.1045237

National Aeronautics and Space Administration (NASA). (2021). Health & Air Quality: 2021 Annual Summary. Washington, D.C.

National Research Council (NRC). (2014). Convergence: Facilitating transdisciplinary integration of life sciences, physical sciences, engineering, and beyond. The National Academies Press.

National Science Foundation (NSF). (2023). Dear colleague letter: Directorate for geosciences (GEO) opportunity for graduate students supplemental funding to link geosciences and human health (GeoHealth INTERN). Retrieved from https://www.nsf.gov/pubs/2023/nsf23112/ nsf23112.jsp

Ni, F. J., Visha, A., Bhavsar, S. P., Arnillas Alberto, C., & Arhonditsis, G. B. (2021). Is it safe to eat fish from the Great Lakes? An adaptive modelling-monitoring framework to assess compliance with consumption advisories. *Journal of Great Lakes Research*, 47(4), 1097–1116. https://doi.org/10.1016/j.jglr.2021.03.016

Ragain, L. (2009). Risk communication and media coverage of emerging contaminants. Journal of the American Water Works Association, 101(5), 100–105. https://doi.org/10.1002/j.1551-8833.2009.tb09892.x

Rice, G. E., Hammitt, J. K., & Evans, J. S. (2010). A probabilistic characterization of the health benefits of reducing methyl mercury intake in the United States. *Environmental Science & Technology*, 44(13), 5216–5224. https://doi.org/10.1021/es903359u

Riverkeeper Inc. (2019). Riverkeeper statement regarding the Champlain Hudson power express. Retrieved from https://www.riverkeeper.org/ news-events/news/energy/riverkeeper-statement-regarding-the-champlain-hudson-power-express/

Rose, D. C., Mukherjee, N., Simmons, B. I., Tew, E. R., Robertson, R. J., Vadrot, A. B. M., et al. (2020). Policy windows for the environment: Tips for improving the uptake of scientific knowledge. *Environmental Science & Policy*, 113, 47–54. https://doi.org/10.1016/j.envsci.2017.07.013

Ruxton, C. H. S. (2011). The benefits of fish consumption. *Nutrition Bulletin*, 36(1), 6–19. https://doi.org/10.1111/j.1467-3010.2010.01869.x

Seitter, K., Tipton, E., Higgins, P., White, L., & Miller, A. (2022). AMS community synthesis on geohealth. The American Meteorological Society. Shimshack, J. P., Ward, M. B., & Beatty, T. K. M. (2007). Mercury advisories: Information, education, and fish consumption. Journal of Environmental Economics and Management, 53(2), 158–179. https://doi.org/10.1016/j.jeem.2006.10.002

Shuchman, M., & Wilkes, M. S. (1997). Medical scientists and health news reporting: A case of miscommunication. Annals of Internal Medicine, 126(12), 976–982. https://doi.org/10.7326/0003-4819-126-12-199706150-00008

Stern, A. H., & Korn, L. R. (2011). An Approach for quantitatively balancing methylmercury risk and omega-3 benefit in fish consumption advisories. *Environmental Health Perspectives*, 111(8), 1043–1046. https://doi.org/10.1289/ehp1002824

Stone, D., & Hope, B. K. (2010). Carcinogenic risk as the basis for fish advisories: A critique. Integrated Environmental Assessment and Management, 6(1), 180–183. https://doi.org/10.1897/IEAM\_2009-002.1

Suleski, J., & Ibaraki, M. (2009). Scientists are talking, but mostly to each other: A quantitative analysis of research represented in mass media. Public Understanding of Science, 19(1), 115–125. https://doi.org/10.1177/0963662508096776

Sundseth, K., Pacyna, J. M., Banel, A., Pacyna, E. G., & Rautio, A. (2015). Climate change impacts on environmental and human exposure to mercury in the Arctic. *International Journal of Environmental Research and Public Health*, 12(4), 3579–3599. https://doi.org/10.3390/ ijerph120403579

Tallis, H., Kreis, K., Olander, L., Ringler, C., Ameyaw, D., Borsuk, M. E., et al. (2019). Aligning evidence generation and use across health, development, and environment. *Current Opinion in Environmental Sustainability*, 39, 81–93. https://doi.org/10.1016/j.cosust.2019.09.004

Tan, M. L., Ujihara, A., Kent, L., & Hendrickson, I. (2011). Communicating fish consumption advisories in California: What works, what doesn't. *Risk Analysis*, 31(7), 1095–1106. https://doi.org/10.1111/j.1539-6924.2010.01559.x

Tilt, B., & Qing, X. (2010). Media coverage of environmental pollution in the People's Republic of China: Responsibility, cover-up and state control. *Media, Culture & Society*, 32(2), 225–245. https://doi.org/10.1177/0163443709355608

United States Environmental Protection Agency (U.S. EPA). (2000). Risk assessment and fish consumption limits (Vol. 2). Office of Science and Technology Office of Water.

United States Environmental Protection Agency (U.S. EPA). (2023). Heavy metals in cultural products. Retrieved from https://www.epa.gov/ children/heavy-metals-cultural-products

United States Food and Drug Administration (U.S. FDA). (2014). Quantitative assessment of the net effects on fetal neurodevelopment from eating commercial fish (as measured by IQ and also by early age verbal development in children).

Van Oostdam, J., Gilman, A., Dewailly, E., Usher, P., Wheatley, B., Kuhnlein, H., et al. (1999). Human health implications of environmental contaminants in Arctic Canada: A review. Science of the Total Environment, 230(1–3), 1–82. https://doi.org/10.1016/s0048-9697(99)00036-4

Wang, Z.-H. (2021). Compound environmental impact of urban mitigation strategies: Co-benefits, trade-offs, and unintended consequence. Sustainable Cities and Society, 75, 103284. https://doi.org/10.1016/j.scs.2021.103284

Wheatley, B., & Paradis, S. (1996). Balancing human exposure, risk and reality: Questions raised by the Canadian aboriginal methylmercury program. *NeuroToxicology*, 17(1), 241–250.

Wheeler, S. A., Carmody, E., Grafton, R. Q., Kingsford, R. T., & Zuo, A. (2020). The rebound effect on water extraction from subsidising irrigation infrastructure in Australia. *Resources, Conservation and Recycling, 159*, 104755. https://doi.org/10.1016/j.resconrec.2020.104755 Willson, M. (2023). *How a \$6B transmission project made it in New York.* E&E News.

Woodall, L. C., Talma, S., Steeds, O., Stefanoudis, P., Jeremie-Muzungaile, M. M., & de Comarmond, A. (2021). Co-development, co-production and co-dissemination of scientific research: A case study to demonstrate mutual benefits. *Biology Letters*, 17(4), 20200699. https://doi. org/10.1098/rsbl.2020.0699

Zhang, J., Li, Y., Tao, W., Liu, J., Levinson, R., Mohegh, A., & Ban-Weiss, G. (2019). Investigating the urban air quality effects of cool walls and cool roofs in southern California. *Environmental Science & Technology*, 53(13), 7532–7542. https://doi.org/10.1021/acs.est.9b00626

Zhang, Y., Bash, J. O., Roselle, S. J., Shatas, A., Repinsky, A., Mathur, R., et al. (2020). Unexpected air quality impacts from implementation of green infrastructure in urban environments: A Kansas City case study. *Science of the Total Environment*, 744, 140960. https://doi. org/10.1016/j.scitotenv.2020.140960